Author's response to reviews

Title: The Relationship of Bottle Feeding and Other Sucking Behaviors with Speech disorder in Patagonian Preschoolers

Authors:

- Clarita Barbosa (pediatria@rehabilitamos.org)
- Sandra Vasquez (svasquezguerrero@yahoo.com)
- Mary A Parada (mary8704@gmail.com)
- Juan Carlos Velez Gonzalez (jcvelez@rehabilitamos.org)
- Chanaye Jackson (msserenity04@yahoo.com)
- N. David Yanez (yanez@u.washington.edu)
- Bizu Gelaye (mirt@u.washington.edu)
- Annette L. Fitzpatrick (fitzpal@u.washington.edu)

Version: 3 Date: 23 June 2009

Author's response to reviews: see over
RE: Manuscript ID: 1077100494283290

Dear Ms. Pafitis:

Thank you for your message and primary review of our manuscript entitled "The Relationship of Bottle Feeding and Other Sucking Behaviors with Speech disorder in Patagonian Preschoolers."

We are enclosing a revised version of our manuscript. Below we outline the modifications we have made in response to your quires.

(1) Please include a 'Background' section in the abstract of your manuscript which includes context information in addition to the aims of your study.
Thank you, the abstract section is now properly structured.

(2) Please include a 'Competing Interests'
We now have included a Competing Interests section.

(3) Please include an Authors’ contributions section
We now have added Authors’ contributions section.

(4) Please provide a copy of the questionnaire administered in your study
We'll make the copy of the questionnaire available up on request.

Please let me know if you have any question.

Thank you for your attention to our manuscript.

Best Regards

Bizu Gelaye
Researcher and Program Manager
Multidisciplinary International Research Training (MIRT) Program
University of Washington, School of Public Health